\documentclass{article}
\usepackage[T1]{fontenc}
\usepackage[a4paper, margin=0.75in]{geometry}

% set up hyperlink coloring
\usepackage{hyperref}
\hypersetup{
    colorlinks=true,
    linkcolor=blue,
    filecolor=magenta,
    urlcolor=cyan
}

\setlength{\parindent}{0pt} % no indentation

% checkbox list: https://tex.stackexchange.com/questions/247681/how-to-create-checkbox-todo-list
\usepackage{enumitem,amssymb}
\newlist{boxlist}{itemize}{2}
\setlist[boxlist]{label=$\square$}

% code support: listings and fancy verbatim
\usepackage{listings}
\lstset{basicstyle=\ttfamily\footnotesize}
\usepackage{fancyvrb}

\renewcommand{\familydefault}{\sfdefault} % default to sans serif

\usepackage{xcolor}
\newcommand{\code}[1]{\colorbox{lightgray}{\texttt{#1}}}
\newcommand{\sh}[1]{\code{\$ #1}}

\usepackage{graphicx}

%%%%%%%%%%%%%%%%
\begin{document}

\title{Advanced Extra}
\author{EECS 201 Fall 2020}
\date{}
\maketitle

\section*{Submission Instructions}
This assignment is an ``online assignment'' on \href{https://gradescope.com}{Gradescope},
where you will fill out a text field.

\section{How's it going? (10)}
Write a paragraph or two (or more) about your experience during this Fall semester.
Feel free to vent about this semester, your classes (including this one),
the election, or whatever else is on your mind.

\end{document}
